

MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING

ALGORITMOS DE APRENDIZADO DE MÁQUINA NA AGRICULTURA: UMA REVISÃO DA LITERATURA SOBRE PREVISÃO CLIMÁTICA E DE PREÇOS, DETECÇÃO DE PRAGAS E DOENÇAS E MONITORAMENTO DE PRODUÇÃO

ALGORITMOS DE APRENDIZAJE AUTOMÁTICO EN LA AGRICULTURA: UNA REVISIÓN DE LA LITERATURA SOBRE PREDICCIÓN CLIMÁTICA Y DE PRECIOS, DETECCIÓN DE PLAGAS Y ENFERMEDADES Y MONITOREO DE PRODUCCIÓN

Emiliano Soares Monteiro¹, Rodrigo da Rosa Righi², Antônio Marcos Alberti³, Sandro José Rigo⁴, Jorge Luis Victoria Barbosa⁵, Perla Haydee da Silva⁶, Lidia Martins da Silva⁷

e626211 https://doi.org/10.47820/recima21.v6i2.6211

RECEIVED: 01/03/2025 ABSTRACT APPROVED: 02/03/2025

PUBLISHED: 02/18/2025

The growing demand for food necessitates the application of advanced technologies in agriculture. Furthermore, due to food production, pests, and climate change incidents are a real-time challenge for farmers. Due to the growing need to apply algorithms in the field, we investigate the algorithms most cited, used, and ongoing projects in the last three years, from 2019 to 2021 Therefore, we evaluated articles that focus on supervised and unsupervised algorithms. This literature review presents an overview of algorithms usage in agriculture. A total of 81 articles were analysed. Our contributions as

¹ Universidade do Estado de Mato Grosso - UNEMAT.

² Professor and researcher at University of Vale do Rio dos Sinos (Unisinos), Brazil. Post-doctoral studies at KAIST - Korean Advanced Institute of Science and Technology, South Korea, under the following topics: RFID and cloud computing. MS and PhD degrees in Computer Science from the Federal University of Rio Grande do Sul, Brazil. Performed a stage at Tecnische Universitaet Berlin, Germany. Member of both IEEE Computer Society and ACM.

³ Associate Professor in Software Engineering at the School of Computer Science, University of Leeds, UK. PhD at the State University of Campinas (Unicamp), Brazil, Researcher at the Electronics and Telecommunications Research Institute (ETRI), South Korea.

⁴ Professor/Researcher in the Applied Computing Graduate Program Program – UNISINOS; Head of the Undergraduate Course in Computer Science in UNISINOS; Head of the Graduate Degree Program in Applied Computer Science in UNISINOS; Dean of the UNISINOS Polytechnic School. BSc in Computer Science at Pontificia Universidade Católica do Rio Grande do Sul PUCRS; MSc in Computer Science at Universidade Federal do Rio Grande do Sul UFRGS; Ph.D. in Computer Science at Universidade Federal do Rio Grande do Sul UFRGS. Professor and Researcher at UNISINOS University. Scholarship for Productivity in Technological Development and Innovative Extension (DT-2) of the National Council for Scientific and Technological Development (CNPq). Main research areas: Artificial Intelligence, Semantic Web, Natural Language Processing, Robotic, Distance Education. Doctoral thesis in progress: 2; Software registered (INPI – Instituto Nacional de Propriedade Intelectual / National Institute of Intelectual Property).

⁵ Graduação em Tecnologia em Processamento de Dados e Engenharia Elétrica pela Universidade Católica de Pelotas (UCPel). Especialização em Engenharia de Software (UCPel). Mestrado e doutorado em ciência da computação na Universidade Federal do Rio Grande do Sul (UFRGS). Pós-doutorado na Sungkyunkwan University (SKKU, Suwon, Coréia do Sul). Pós-doutorado na University of California Irvine (UCI, Irvine, USA) através de uma bolsa do Programa CAPES/PRINT (professor visitante no Exterior Sênior). Professor Titular II na Universidade do Vale do Rio dos Sinos (Unisinos). Professor no Programa de Pós-graduação em Computação Aplicada (PPGCA) e no Programa de Pós-Graduação em Engenharia Elétrica (PPGEL). Coordenador do Laboratório de Computação Móvel (Mobilab) e atua como Bolsista de Produtividade em Desenvolvimento Tecnológico e Extensão Inovadora (bolsa DT - atualmente no Nível 1C) do Conselho Nacional de Desenvolvimento Científico e Tecnológico (CNPq).

⁶ PhD in Linguistic Studies from the Institute of Languages at UFMT (PPGEL). Holds a Master's degree in Applied Linguistics from the Institute of Languages at UFMT (PPGEL), a specialization in Higher Education Didactics from UNIC - Universidade de Cuiabá, a Bachelor's degree in Law from UNIC - Universidade de Cuiabá, and a Bachelor's degree in Language and Literature from UFMT - Universidade Federal de Mato Grosso.

⁷ Doutorado em Computação Aplicada pela Universidade do Vale do Rio dos Sinos Unisinos, Mestrado em Ciência da Computação, Especialização em Formação Tecnológica e Inclusiva, Especialização em Informática na Educação, Especialização em Design Instrucional, Graduação em TPD - Tecnologia em Processamento de Dados. Coordenadora dos Laboratórios de Informática e do Núcleo de Processamento de Dados do Centro Universitário Cândido Rondon, Professora do Curso Ciência da Computação e Tecnologia em Análise de Sistemas do Centro Universitário Cândido Rondon. Tutora presencial na Centro Universitário Claretiano, Tutora no IFMT, professora e tutora nos cursos a distância da UFMT, Professora no curso de Pós Graduação Governança de TI do SENAC- MT, professora no curso de Sistemas da Informação no UNIVAG, Coordenadora dos Cursos: Tecnologia em Análise e Desenvolvimento de Sistemas e Bacharelado em Ciência da Computação na Faculdade INVEST de Ciências e Tecnologia.



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

a) an analysis of the state-of-the-art on applying algorithms to various agricultural functions and b) a taxonomy to help researchers, governments, and farmers choose these algorithms. The literature review systematization methodologically summarizes the following: the definition of research objectives and questions, the definition of criteria, search strategies, screening of results, synthesis, and analysis of data, discussion, review, and conclusions. This article adds discoveries about the application of algorithms in crops, machinery, and processes and points out new lines of research.

KEYWORDS: Algorithm. Agriculture. Random Forest. Machine Learning. Drought. Forecast. Al.

RESUMO

A crescente demanda por alimentos exige a aplicação de tecnologias avançadas na agricultura. Além disso, devido à produção de alimentos, as pragas e os incidentes de mudanças climáticas são um desafio em tempo real para os agricultores. Devido à crescente necessidade de aplicação de algoritmos no campo, investigamos os algoritmos mais citados, utilizados e projetos em andamento nos últimos três anos, de 2019 a 2021 Portanto, avaliamos artigos que enfocam algoritmos supervisionados e não supervisionados. Esta revisão da literatura apresenta uma visão geral do uso de algoritmos na agricultura. Foram analisados 81 artigos. Nossas contribuições são a) uma análise do estado da arte na aplicação de algoritmos a várias funções agrícolas e b) uma taxonomia para ajudar pesquisadores, governos e agricultores a escolher esses algoritmos. A sistematização da revisão de literatura sintetiza metodologicamente: a definição dos objetivos e questões da pesquisa, a definição de critérios, estratégias de busca, triagem de resultados, síntese e análise dos dados, discussão, revisão e conclusões. Este artigo acrescenta descobertas sobre a aplicação de algoritmos em culturas, máquinas e processos e aponta novas linhas de pesquisa.

PALAVRAS-CHAVE: Algoritmo. Agricultura. Floresta aleatória. Aprendizado de máquina. Seca. Previsão. IA.

RESUMEN

La demanda de alimentos está creciendo cada año y exige aplicaciones tecnológicas más significativas en el campo. Además, debido a la producción de alimentos, las plagas y los incidentes relacionados con el cambio climático representan un desafío en tiempo real para los agricultores. Ante la creciente necesidad de aplicar algoritmos en el campo, investigamos los algoritmos más citados, utilizados y proyectos en curso en los últimos tres años, de 2019 a 2021. Por lo tanto, evaluamos artículos cuyo enfoque principal fue en algoritmos de aprendizaje supervisado. Esta revisión de literatura presenta una visión general del uso de algoritmos en la agricultura. Se analizaron un total de 81 artículos. Nuestras contribuciones incluyen: a) un análisis del estado del arte en la aplicación de algoritmos a diversas funciones agrícolas y b) una taxonomía para ayudar a investigadores, gobiernos y agricultores a elegir estos algoritmos. Este artículo aporta descubrimientos sobre la aplicación de algoritmos en cultivos, maquinaria y procesos, además de señalar nuevas líneas de investigación.

PALABRAS CLAVE: Algoritmo. Agricultura. Bosque aleatorio. Aprendizaje automático. Sequía. Previsión. IA.

1. INTRODUCTION

Agriculture is essential for developing cultures that, in addition to feeding the nation, are the base for exportation from several third world countries. In many of these countries, agriculture represents a large share of the resources coming from exports to the national economy (Wang *et al.* 2019). Due to the importance of agriculture for the local and export economy, several technologies are used to maintain, and control products such as the use of drones (Chinnaiyan; Balachandar,



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

2020), IoT (Garg *et al.*, 2021), satellite imagery (Moumni; Lahrouni 2021) and mostly machine learning.

The primary needs of farmers have continued to be problematic addressed in various corners of the world for a long time, such as combating various types of pests (fungi, bacteria, insects, and rodents); water resources management trying to maintain the water supply and avoid drought; knowledge the players in the production chain and maintain trusting relationships within this supply chain; predict climate changes that affect crops and crops; price forecasting is a problem that affects futures contracts for the sale of grains.

Problems such as droughts, pests, uncertainties about price determination, the need for planning and monitoring, the use of technologies are addressed as some challenges for agriculture, and the relationship of algorithms in the treatment of these issues. Figure 1. There is a gap on identifying how machine learning is being used in the last years, this literature review focuses on this time window. This work presents contributions that serve for future research; we highlight current research on machine learning, present a simplified taxonomy of the application of machine learning in agriculture, and indicate future opportunities and challenges in this area.



Figure 1. Challenges for agriculture

The text is organized as follows: Section 2 presents the materials and methods and research questions; section 3 presents the results obtained from the selected articles; section 4 presents the limitations; section 5 presents the challenges in the area with open questions; we end in section 6 with our conclusions and indications for future work.



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

2. MATERIALS AND METHODS

The current study is a systematic literature review designed to provide a broad view of the application of machine learning algorithms in agriculture. The Biolchini *et al.*, (2005) and Budgen and Brereton (2006) methodology was followed to outline a plan for this systematic literature review. This review was carried out following these steps: Planning; development of research questions; description of information sources and operationalization of data collection; description add from data sources; description of choice criteria; description of quality control and refinement criteria; comparison of filter results against search questions.

2.1. Planning

A protocol was developed that included: search sources, filtering criteria, and comparisons between filter results and search questions related to the application of algorithms in agriculture. This literature review attempted to identify unique studies after applying inclusion and exclusion criteria. These were summarized in the results. One author developed the filtering process, applied criteria, and research questions; other authors reviewed the research questions and carried out reviews. Online electronic databases were used for the searches (cited below in the section: Information sources).

2.2. Research questions

The questions for the survey were defined as they are important to define the direction of the review process. With the questions, the authors intend to identify cutting-edge and innovative research that focuses on the application of machine learning algorithms in agriculture; We also aim to identify new trends, challenges, and solutions with the questions. The Table 1 describes the questions.

ld	Question
RQ1	Which studies use algorithms for forecasting whether or drought?
RQ2	What algorithms are used for pest, disease, or deficiency detection?
RQ3	What algorithms are used for price prediction?
RQ4	What algorithms are used in planning, controlling, monitoring agricultural
	production?
RQ5	What technologies are used in conjunction with machine learning?
RQ6	Is it possible to elaborate on a basic taxonomy?
RQ7	What other bibliometric data can be obtained?

Table 1. Research questi	ions
--------------------------	------



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

2.3. Information sources

Online electronic databases were selected that allow a broad search in the current literature to collect a more significant number of articles. Online electronic databases were: ACM, Hindawai, IEEE, MDPI, Microsoft Academic, PlosOne, ScienceDirect, Springer, Taylor and Francis Online.

2.4. Inclusion and exclusions criteria

After all, studies were obtained, the inclusion and exclusion process were started. This process is a filter on the obtained jobs. The most relevant articles relating to the topic were selected by using the criteria. Also removed are conference names correlated with the search strings due to the particularities of each electronic database. Articles are separated by titles and abstracts and then grouped. In this grouping, duplicate articles based on title are removed, and later we removed the duplication based on authors, as some works repeatedly appear in more than one database. The article's overview approach is carried out to detect particularities in the text as specific technologies and algorithms. The number of articles removed is significant due to the difficulty of distinguishing between the application of statistical methods, other surveys, or very small abstracts that still appear; non-aligned articles are removed. Only articles in English were selected from 2019 to 2021. The period chosen between 2019 and 2021 is important because it addresses the pre-pandemic and alternatives without the influence of LLM (large language model) algorithms as launched such as ChapGPT launched in 2022, in this way this article is a time frame allowing comparability of the algorithms used before this period, in addition to not having identified previous similar reviews, and thus created the possibility of post comparison of prompt-based tools and reanalysis of these other algorithms. The criteria are in Table 2.

Id	Inclusion criteria
IC1	Articles in English
IC2	Publication period: 2019 to 2021
IC3	Articles containing the search strings in the title
ld	Exclusion criteria
EC1	Articles older than 3 years
EC2	Complete books
EC3	Language other than English
EC4	Theses, dissertations
EC5	Duplicate publications

Table 2. Inclusion and exclusion criteria



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

EC6	Commercial materials or white paper
EC7	Publication with less than 4 pages
EC8	Articles with anonymous authors
EC9	Literature reviews or surveys
EC10	Corrections or errata
EC11	Academic monographs
EC12	Regulations and policies

2.5. Quality assessment

We adopt some quality criteria during the screening selection process that are important for the general analysis of the collected works. Among them, we cite contextualization of work with current problems, citation of technologies of interest to this search, if the article presents a machine learning application to solve problems if the article presents future work or indicates directions for future improvements.

3. RESULTS

The results are presented in the following section. The answers to the questions are presented with the help of the general interpretation of the selected works. The applied article selection process found 6,030 articles, of which, after passing through the filter funnel, resulted in 88 articles used. Articles were grouped according to the publisher: ACM 5, arXiv 2, BASE 2, Hindawi 3, IEEE 11, MDPI 26, Microsoft Academic 4, PlosOne 2, ScienceDirect 31, Springer 1, Tandfonline 2.



Figure 2 show publications grouped by year (according to the inclusion criteria).

Figure 2. Publications per year



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

3.1. RQ1 - Which studies use algorithms for forecasting whether or drought?

The work of Ketchum *et al.*, (2020) developed a land classifier using a geospatial database and LandSat satellite images to detect irrigated areas; the one classified used was the random forest (RF) identifying areas with decreased irrigation and others with increase. The work of Xavier *et al.* (2020) used machine learning (especially the algorithms: (RF) and decision trees (DT) to understand the occurrence of drought in northeastern Brazil, the characterization of drought used the standard precipitation index, among other data; the use of RF and DT allowed farmers to adopt and adapt strategies for water management, especially in water deficit scenarios; In time series analysis there is data delay. Thus, the use of other non-linear techniques were applied.

Muniasamy (2020) They developed a study indicating the applicability of IoT and machine learning in data collection, processing, and decision making in intelligent agriculture, citing the use of robots to monitor plants and pests; cites the use of drones to monitor soil and crops and analytics to predict events based on historical data on temperature, humidity, wind speed, and others. A proposal for a machine learning approach to rainfall forecast, using several attributes such as humidity and temperature applying the algorithms multiple linear regressions (MLR), K nearest neighbor (KNN), support vector machines (SVM), RF; of which the RF had the best accuracy, Premachandra (2021).

Six machine learning models classification and regression trees (CART), boosted regression trees (BRT), random forest (RF), multivariate adaptive regression splines(MARS), flexible discriminant analysis(FDA), and SVM) were used to map drought risks in Australia from 1994 to 2013; producing a risk map that allows the planning of water resources and contingency, Rahmati *et al.*, (2019) Rahmati *et al.*, (2020). Moumni and Lahrouni (2021) carried out identification of types of crops based on classified images via machine learning, using SVM and artificial neural network (ANN), images from the Sentinel-1 and Sentinel-2 satellites showing the soil mapping; demonstrated that applications of these classifiers increase when working with images.

Prodhan *et al.* (2021) aimed to identify and pinpoint the magnitude and duration of future drought occurrences using data from pre-existing designs applied to RF and gradient boosted models (GB). Gong *et al.* (2021) used evapotranspiration data with machine learning algorithms (particle swarm optimization (PSO), genetic algorithm (GA), and extreme learning regression (ELR) to estimate the amount of water during the crop irrigation process. Li *et al.* (2021) develop a disaster vulnerability model for drought risks; also aiming to predict crop yields and insurance rates; to assist in agronomic management, they used satellite images and machine learning, used various data such as precipitation, leaf area, and others; the machine learning algorithms used were RF and multi linear regression (MLR).

Dubois *et al.* (2021) modeling of the irrigation problem was carried out so that predictions could be made, including soil and potato crop data; ANN, RF, and SVM were used; were successful in predicting water values. Feyisa *et al.* (2020) applied time series of vegetation indices data obtained from the Moderate Resolution Imaging Spectrometer (MODIS) with machine learning algorithms



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

(SVM, RF, and C5.0 (C5 algorithm)) to perform irrigation and rainfall patterns. Yamac, and Todorovic (2020) carried out a study on the evapotranspiration of potato plantations using machine learning algorithms (KNN, ANN and adaptive boosting (Adaboost)), indicate the K nearest neighbor (KNN) when meteorological data is limited, in other situations they recommend the use of ANN.

Folberth *et al.* (2019) applied gradient boost and RF to predict the output of models with a low and acceptable spatial-temporal resolution for the corn crop, indicating development potential by applying machine learning in the development of crop model emulators. The articles mentioned above are summarized in Table 3 with data on authors, year, country of the first author, number of citations and main algorithm mentioned in the body of the article; articles related to Research Question 1.

Author	Year	Database	Country	Citations	Algorithm
[Ketchum et al. 2020]	2020	MDPI	USA	6	RF
[Xavier et al. 2020]	2020	MDPI	Brazil	3	RF, DT
[Muniasamy 2020]	2020	IEEE	Saudi		not specified
			Arabia		
[Premachandra and Kumara	2021	IEEE	Sri Lanka	17	SVM, RF, KNN,
2021]					MLR
[Rahmati et al. 2020]	2020	ScienceDirect	Viet Nam	40	SVM, CART, RF,
					BRT,
					MARS, FDA
[Moumni and Lahrouni 2021]	2021	Hindawi	Morocco	5	ML, ANN, SVM
[Prodhan et al. 2021]	2021	ScienceDirect	China		RF, GBRT
[Gong et al. 2021]	2021	ScienceDirect	China	6	ELR, PSO, GA
[Li et al. 2021]	2021	ScienceDirect	China	2	RF, MLR
[Dubois et al. 2021]	2021	ScienceDirect	France	3	ANN, SVM, RF, DT
[Feyisa et al. 2020]	2020	ScienceDirect	Ethiopia	5	C5.0, SVM, RF
[Yamac, and Todorovic 2020]	2020	ScienceDirect	Turkey		AB, ANN, KNN
[Folberth et al. 2019]	2019	ScienceDirect	Austria	48	RF, GBRT

Table 3. Authors, year, electronic database, citations, algorithm

3.2. RQ2 - What algorithms are used for pest, disease or deficiency detection?

Garg *et al.* (2021) used IoT to collect data and apply it to various machine learning algorithms in precision agriculture; among the algorithms used are: RF, light gradient boosting machine (LGBN),



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

XGBoost (XGB), decision tree (DT), KNN and convolutional neural network(CNN), the data used were several, among them irrigation and fertilization aiming to perform the prediction of damage in crops. Shukla *et al.* (2021) Shukla *et al.* (2021) proposed a harvest monitoring system that manages data and devices and supports data sharing and communication by combining IoT and unmanned aerial vehicle (UAV); the combination of IoT and machine learning K-means (KM), ensured the accuracy in predicting harvest diseases.

Agrawal *et al.*, (2021) used machine learning to detect diseases in plantations, the algorithms used were naive bayes (NB) and neural networks (NN). Beck *et al.* (2020) developed a robotic system that captures plants and uses machine learning to identify them from various angles; they could separate and label plants; remove the background and seed images; the authors used CNN. Towett *et al.*, (2020) used machine learning (XGBoost) to quantify the concentrations of nutrient elements (macro and micro) in soil samples and propose a quality control to improve the quality of soil physical condition via organic amendments.

Via machine learning application an ANN and SVM to detect diseases in potato plantations; the experiment was carried out in rural Italy; climatic variables used in the study came from meteorological stations, SVM was more efficient than ANN, Fenu and Malloci (2019). Using Sentinel-2 satellite images, the PLS, RF, and gradient boosted (GB) algorithms were applied to analyze estimates of leaf area and chlorophyll content; RF proved to be better and applicable in crop monitoring, Kganyago *et al.* (2021).

latrou *et al.* (2021) carried out the mapping of nitrogen demand in rice planting using XGBoost, treating satellite images (Sentinel 1 and 2), aiming to predict the growth stage. Machine learning algorithms (ANN, RF, and multiple linear regressions (MLR)) was applied to satellite images (Terra MODIS) to identify the insect Nilaparvata lugens, a pest that attacks rice; data crossed with meteorological and satellite information proved to be correct in detecting the pest and developing an alert system, Skawsang *et al.* (2019).

Manrique-Silupu *et al.* (2021) used IoT sensors in banana plantations to estimate pest classification based on machine learning techniques via SVM processing atmospheric data; obtained satisfactory results in the application of spray-on more effective dates. Through the collection of infrared wave data captured in hyperspectral carried out in overflights in corn plantations, these data collected were applied partial least squares (PLS) and RF; indicating the possibility of using data combined with machine learning techniques to quantify planting trails for precision agriculture, Wang *et al.* (2021).

Balakrishna *et al.* (2021) approach realization of the union of IoT with machine learning to detect the invasion of animals in the fields where the plantations are located; used Raspberry PI and cameras with the CNN algorithm to detect animals; detection is sent via Twilio to farmers. Dash *et al.* (2021) approach the establishment of the relationship between micronutrients and macronutrients and



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

climate parameters to classify the crops (rice, wheat, and sugarcane) applied machine learning algorithms SVM and DT.

Danner *et al.* (2021) developed biophysical and biochemical data processing with machine learning algorithms (ANN, RF, SVR, and gaussian process regressions (GPR)) by spectroscopic image analysis; ANN was considered more efficient. Kasinathan et al. (2021) developed an algorithm for detecting insect pests using the dataset (Wang, Xie, Deng, and IP102) through various algorithms (ANN, SVM, KNN, NB, and CNN), CNN proved more efficient.

Hu *et al.* (2020) to identify, control, and predict the levels of heavy metals in crop soils, they used RF and gradient boost (GB) algorithms with variables of soil and crop chemical properties; the authors were able to predict the presence of heavy metals in harvests and reduce time and cost of laboratory analysis of samples. Radhakrishnan (2020) applied the combination of VMS with CNN together to detect the incidence of diseases in rice caused by fungi and bacteria.

The above-mentioned articles are summarized in Table 4 with data from authors, year, parents of the first author, number of citations and main algorithm mentioned in the body of the article; articles related to research question 2.

Author	Year	Database	Country	Citations	Algorithm
[Garg <i>et al.</i> 2021]	2021	arXiv	India		RF, LGBM, DT,
					KNN, CNN, XGB
[Shukla <i>et al.</i> 2021]	2021	Microsoft	India		КМ
		Academic			
[Agrawal <i>et al</i> . 2021]	2021	IEEE	India	87	ANN, NB
[Beck et al. 2020]	2020	Plos One	Canada	3	CNN
[Towett et al. 2020]	2020	Plos One	Kenya		XGB
[Fenu and Malloci 2019]	2019	ACM	USA		ANN, SVM
[Kganyago <i>et al</i> . 2021]	2021	MDPI	South		PLS, RF, GBRT
			Africa		
[latrou <i>et al</i> . 2021]	2021	MDPI	Greece	1	XGB
[Skawsang et al. 2019]	2019	MDPI	Thailand	7	ANN, RF, MLR
[Manrique-Silupu et al.	2021	ScienceDirect	Peru		SVM
2021]					
[Wang et al. 2021]	2021	ScienceDirect	USA		RF

Table 4. Authors, year, electronic database, citations, algorithm



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

[Balakrishna <i>et al.</i> 2021]	2021	ScienceDirect	India		CNN
[Dash <i>et al</i> . 2021]	2021	ScienceDirect	India		SVM, DT
[Danner <i>et al.</i> 2021]	2021	ScienceDirect	Germany	8	GPR, ANN, SVR, RF
[Kasinathan <i>et al.</i> 2021]	2021	ScienceDirect	India	4	ANN, SVM, KNN, CNN, NB
[Hu <i>et al.</i> 2020]	2020	ScienceDirect	China	41	RF, GBRT,
[Radhakrishnan 2020]	2020	ScienceDirect	India	2	SVM, CNN

3.3. RQ3 - What algorithms are used for price prediction?

Yuan *et al.* (2020) proposed a model for commodity price prediction so that government agencies could promote new policies and farmers to improve crop planning. They have developed an algorithm with lag time inspired by a biological evolutionary differential to perform the selection of ideal time for price prediction; the focus is the plantations of Malaysia; (ANN, particle swarm optimization (PSO), and genetic algorithms (GA)).

Ahish *et al.* (2019) analyzed historical data with several regression models (linear, multilinear, and nonlinear) proposing a predictive model for price, which helps in the analysis of the viability of the harvest according to market prices, generating final reports with a graphical representation; (DT, autoregressive integrated moving average (ARIMA)). Chen *et al.* (2021) developed a system for predicting agricultural commodity prices by putting up a web system, where they evaluated several algorithms identifying LST as the best; they applied research in Malaysian market price data; used autoregressive integrated moving average (ARIMA), support vector regression (SVR), Prophet, long short-term memory model (LSTMM), XGBoost.

The above-mentioned articles are summarized in Table 5 with data from authors, year, parents of the first author, number of citations and main algorithm mentioned in the body of the article; articles related to research question 3.

Author			Year	Database	Country	Citations	Algorithm
[Yuan	et	al.	2020	ACM	Malaysia		ANN, PSO, GA
2020]							

Table 5. Authors, year, electronic database, citations, algorithm



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

[N et al. 2019]	2019	BASE	India		DT, ARIMA
[Chen <i>et al.</i> 2021]	2021	arXiv	Malaysia	1	ARIMA, SVR, Prophet, XGBoost, LSTMM

3.4. RQ4 - What algorithms are used in planning, controlling, and monitoring agricultural production?

Wang *et al.* (2019) proposed an intelligent evaluation method based on machine learning using backpropagation (BP) neural network and support vector machine (SVM) algorithms to establish a high-tech agricultural evaluation index. They indicate that in the area of evaluation of high-tech agriculture, quantitative indices are used, it proposes methods of qualitative analysis according to the experiences of experts in the area using neural networks.

Chaterji *et al.* (2021) collected data through a vast network of IoT devices; presented the Lattice platform, which integrates the IoT devices, processes the data, stores statistics, and presents information about the collected data; used density based spatial clustering of applications with noise (DBSCAN), ANN and Bayesian models (BM). Peppes *et al.* (2021) considers an increase in network attacks on IoT devices and sensors arranged in networks, so you proposed the use of machine learning to perform traffic analysis (employing various algorithms such as KNN, support vector classification (SVC), DT, RF and stochastic descent gradient (SGD)).

To perform the identification of machinery operating in a field, several data were collected about the operation of the machine, seeking to identify which machine works in a given field, the algorithms KNN, SVM, DT, RF, GB were used; the RF showed better results in the classification of machinery, Waleed *et al.* (2021). The authors developed a method that allows the mapping of carrot culture and its yield. The algorithm used was RF, processing data from a satellite spectral database, Wei et al. (2020).

The authors discussed an application of a machine learning algorithm (ID3) to develop an agricultural consultation system (AAS), focusing mainly on soil degradation and crop production, being a recommendation system, Bhimanpallewar and Narasingarao (2020). Concerned about work accidents in the field, the authors Scott *et al.* (2021) used naive Bayes (NB) to identify injuries using an administrative database and pre-hospital data; reduced cases of visual inspection.

Balne (2020) using climate and soil information, inserted in a framework developed as an Android application, this application works together with Multi Linear Regressions (MLR) to indicate the best crop when given soil and climatic conditions. Feng *et al.* (2019) determined that remotely perceived factors contribute to monitoring drought in agriculture (in south-east Australia), using rain



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

and satellite data; three machine learning methods were used RF, SVM, and multi-layer perceptron neural network (MLPNN); RF was the one that showed the best results.

Jagtap *et al.* (2021) presented the applicability of several methods of machine learning in agriculture, among them KNN, Bayesian model (BM), ANN, CNN, DT, RF, and SVM), indicating the possibility of economic applicability in various areas of agriculture. The authors Mouhssine *et al.* (2020) proposed to use machine learning to classify red fruits from a farm without human intervention were used KNN, deep neural networks, and CNN.

Muthoni *et al.* (2021) estimated temporal variations in maize grains over 13 years in four African countries; applying RF on several variables (such as rain and humidity) could detect variations in the grains; indicate that data from various sources coupled with machine learning algorithms can be used to predict actions and aid agriculture. Liu and Zhan (2019) studied the financial efficiency of listed Chinese agricultural companies using RF machine learning and the Tobit regression model; RF stands out; the authors point out which indicators improve the performance of companies.

Adebiyi *et al.* (2020) developed a mobile support system that uses machine learning to optimize land use; this system receives various input data such as geolocation, soil type, and crop; using RF classifies cultures according to their application, used solutions to develop such as BigML e Appery.io. Tang *et al.* (2020) proposed a multiscale object-based weighted method classification method with manual digitization of harvest distribution in Gaofeng-2 satellite images.

Sosa *et al.* (2021) proposed the assessment of hail damage in crops using machine learning with KM clustering pixel of damaged areas in Sentinel1 and 2 satellite images; applied the study to corn, soybeans, and wheat. Acharya *et al.* (2021) conducted studies with machine learning algorithms to predict soil mix, including CART, RF, BRT, MLR, SVR, and ANN; the authors indicate better performance for RF and BRT.

Banerjee *et al.* (2021) developed methods to estimate wheat seedlings using multispectral images captured from unmanned aerial vehicles (drones); the algorithms were regression trees, support vector, and Gaussian regression; show promising results at 10 m altitude with 20-day seedlings. Crisostomo de Castro Filho *et al.* (2020) carried out the mapping of rice plantations via satellite images (Sentinel-1) processed via machine learning with ANN and LSTM and compared with SVM, RF, KNN, and NB; a bidirectional long short-term memory (LSTM) proved to be more accurate.

Abbas *et al.* (2020) applied proximal sensing techniques to investigate soil and crops in potato cultivation, including machine learning with KNN, SVR, and linear regression; conclude that a vast and broad database is needed to generate more accurate results for each model that can be applied in specific areas. Ding *et al.* (2020) evaluated machine learning regression machines using multispectral data from Sentinel-2 satellite and unmanned aerial vehicles; partial least squares regression (PLSR), ANN, Gaussian process, SVR, and RF were analyzed; and the recommended approach was Gaussian processes to quantify crop residue coverage.



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

Maimaitijiang *et al.* (2020) the combination of data from satellites (Worldview-2/3) and unmanned aerial vehicles and machine learning was performed; PLSR, RFR, SVR, and ELR algorithms were used; studies focus on soy plantations and indicate that combining diverse image data with machine learning techniques increases crop monitoring opportunities. Machine learning application via gray level cooccurrence matrix (GLCM), RF, and SVM on images captured by unmanned aerial vehicles allowed for crop classification; the results indicate that only one image is used for crop classification, Kwak and Park (2019).

Yang *et al.* (2019) indicates that data from remote sensors can be allocated in grids for crop classification; applying parallel computation techniques increased the computation capacity on the grid; used data from Gaofen and Sentinel satellite images submitted to machine learning with SVM and RF; they end by indicating that parallel computing helps in the data processing. Zhang *et al.* (2019) used KNN, RF, and SVM algorithms under a spectral database aiming at harvest classification; the data crossing also used crop classifiers; created a spectral library combined with the characteristics of cultures allowing for remote monitoring.

Mateo-Sanchis *et al.* (2021) aimed to investigate the agro-ecological influence and regions of influence to estimate harvest using gaussian process regression (GPR), the combination of variables such as soil and vegetation, and their relationship in the model that allows identifying anomalous crops and their failure factors. Tufail *et al.* (2021) used multispectral data from Synthetic Aperture Radar and Multispectral Instruments from Sentinel 1 and 2 satellites with RF algorithms to perform a mapping of crop types and used DT in the study of land use.

Pant *et al.* (2021) a machine learning approach (gradient boosted (GB), RF, SVM, and DT) was used to develop a model capable of identifying patterns in corn, potato, rice, and wheat crop data, seeking to identify patterns for crop prediction. Paudel *et al.* (2021) performed the combination of crop modeling with machine learning (KNN), including climate, soil, and remote sensing data; five crops were studied: wheat, barley, sunflower, sugar cane, and potato), the increase in data indicates greater and better results in the forecasts.

Maponya *et al.* (2020) used satellite images (Sentinel-2) to classify harvest types using the algorithms (SVM, DT, KNN, RF, and ML), indicating that SVM and RF obtained better results than other algorithms. Schwalbert *et al.* (2020) to improve soybean crop forecasts, they used LSTM and RF applied to satellite images and variables such as soil temperature and precipitation, demonstrating the possibility of integrating statistics with remote sensing.

Wolanin *et al.* (201) performed the analysis of satellite images (Landsat 8 and Sentinel-2) applying machine learning with neural networks (ANN) to estimate crop yield for C3 and C4 crops. Tao and You *et al.* (2019) used multi-layer perceptron neural network (MLPNN)) and neural networks applied (ANN) to satellite images to perform prediction of adoption of cover crop adoption for the corn crop.



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

The above-mentioned article sums up in table 6 with data from authors, year, parents of the first author, number of citations and main algorithm mentioned in the body of the article; articles related to research question 4.

Author	Year	Database	Country	Citations	Algorithm
[Wang et al. 2019]	2019	ACM	China		ANN, SVM
[Chaterji et al. 2021]	2021	IEEE	USA	2	DBSCAN, ANN, BM
[Peppes et al. 2021]	2021	MDPI	Greece		KNN, SVC, DT, RF SGD
[Waleed et al. 2021]	2021	MDPI	Korea	2	KNN, SVM, DT, RF, GB
[Wei et al. 2020]	2020	MDPI	Brazil	7	RF
[BhimanpallewarandNarasingarao 2020]	2020	IEEE	India	1	ID3
[Scott et al. 2021]	2021	Springer	USA		NB
[Balne 2020]	2020	Microsoft Academic	India		MLR
[Feng et al. 2019]	2019	ScienceDirect	Australia	57	MLPNN, SVM, RF
[Jagtap et al. 2021]	2021	ScienceDirect	India		SVM, RF, DT, KNN, CNN, BM, KM
[Mouhssine et al. 2020]	2020	Microsoft Academic	Morocoo		KNN, CNN
[Muthoni et al. 2021]	2021	IEEE	Tanzania	31	RF
[Liu and Zhan 2019]	2019	Hindawai	China	9	RF
[Adebiyi et al. 2020]	2020	Hindawai	Nigeria	7	RF
[Tang et al. 2020]	2020	IEEE	China	2	SVM
[Sosa et al. 2021]	2021	MDPI	Spain		КМ

Table 6. Authors, year, electronic database, citations, algorithm



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

[Acharya et al. 2021]	2021	MDPI	USA		ANN, SVR, CART, RF, BRT, MLR
[Banerjee et al. 2021]	2021	MDPI	Australia		GPR, SVM, CART
[Crisostomo de Castro Filho et al. 2020]'	2020	MDPI	Brazil	22	ANN, LSTMM
[Abbas et al. 2020]	2020	MDPI	Canada	17	EN, SVR, KNN
[Ding et al. 2020]	2020	MDPI	China	4	GPR, PLSR, ANN, SVM, RF
[Maimaitijiang et al. 2020]	2020	MDPI	USA	27	PLSR, SVR, ELR, RF
[Kwak and Park 2019]	2019	MDPI	Korea	29	GLCM, SVM, RF
[Yang et al. 2019]	2019	MDPI	China	27	SVM, RF
[Zhang et al. 2019]	2019	MDPI	China	9	SVM, RF, KNN
[Mateo-Sanchis et al. 2021]	2021	ScienceDirect	Spain		GPR
[Tufail et al. 2021b]	2021	ScienceDirect	Pakistan		RF, DT
[Pant et al. 2021]	2021	ScienceDirect	India		SVM, RF, DT, GBRT
[Paudel et al. 2021]	2021	ScienceDirect	Netherlands	4	KNN
[Maponya et al. 2020]	2020	ScienceDirect	South Africa	19	ML, SVM, RF, DT, KNN
[Schwalbert et al. 2020]	2020	ScienceDirect	Brazil	44	RF, LSTMM
[Wolanin et al. 2019]	2019	ScienceDirect	Germany	47	ANN
[Tao and You 2019]	2019	ScienceDirect	USA	1	ANN, MLPNN, RF



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

3.5. RQ5 - What technologies are used in conjunction with machine learning?

Mazzia *et al* (2020) presented a framework for satellite image processing using deep learning techniques and the use of unmanned aerial vehicles; the classifier algorithm used was KM. Abdellah and Thangadurai (2021) developed an IoT framework so farmers could view their data remotely; used the Blynk IoT platform receiving data from Raspberry Pi; allowing them to start the irrigation process, suggest fertilizers, and detect moving objects in the plantation by capturing images.

Gumma *et al.* (2020) used RF to create a Landsat satellite image bank and process the algorithm in the Google Earth Engine cloud; this big data presented a vast picture of Southeast Asia's agricultural production. Tufail *et al.* (2021) developed equipment for the intelligent application of agrochemicals in tobacco plating; the equipment can be installed on a tractor; machine learning used the SVM classifier running on a Raspberry Pi. Shafi *et al.* (2020) present a proposal for the integration of technologies: IoT, machine learning, and Drones to allow the analysis of harvest health; the proposal unifies data collected from IoT sensors with multispectral data collected by drones; the data are later validated via field specialist.

Aneece and Thenkabail (2021) used historical data from Earth Observing-1 (EO-1) with data from the DLR Earth Sensing Imaging Spectrometer via machine learning in Google Earth Engine to classify crops in the USA; the algorithms used were RF, SVM, NB and WekaXMeans; the use of cloud computing associated with machine learning facilitates the analysis and classification in agriculture. Worrall *et al.* (2021) used machine learning via LSTMM and ANN to estimate corn harvest growth, developing the domain guided neural network (DgNN) method that includes domain knowledge.

Yang and Cho (2021) developed a solution for rebuilding 3D images of plants from 2D images captured with Microsoft Kinect V2, applying machine learning algorithms such as DBSCAN and KM; several other algorithms for image treatment were used; the combination of these demonstrated errors below 5mm in the reconstruction of 3D images. Barnes *et al.* (2021) isolated photosynthetic activity signal via satellite images by machine learning application using RF; incorporated thermal data and reflectance metrics.

Najafi *et al.* (2021) performed the comparison of image analysis methods via Fuzzy analysis and methods of machine learning algorithms (SVM and ANN) aiming to analyze the vegetation residue coverage; unmanned vehicles and satellite (Sentinel-2) were used. Xie *et al.* (2021) performed the harvest height estimation with PoISAR and RADARSAT-2 radar images in agricultural areas of Canada; the data were applied to the RF and SVR methods; RF had the best results in obtaining height data from the radar.

Momm *et al.* (2020) Developed a procedure to generate a history of crop types from vegetation index data collected from satellite (Landsat 5); this data was applied evaluated via RF algorithm, the crops studied were corn and soybean. Yan *et al.* (2021) presented a method for merging images of multiple satellites (Sentinel-2 and GaoFen-1) to observe time-space data by



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

applying a grid framework; used machine learning via classification and regression trees (CART) aiming at the classification of crops.

Using the same types of input variables with data on rice, corn, and soybean crops, and seven machine learning algorithms were evaluated (SVM, DT, RF, ANN, stacked sparse autoencoder (SSAE), CNN, and long short-term memory (LSTM)); identifying the SVM as the efficient, Ju *et al.* (2021) [Ju *et al.* 2021]. By using satellite images (Sentinel 1 and 2) to monitor portions of arable area in Germany, RF was used to classify regions for green emissions, data with normalized difference vegetation index (NDVI) were used as input, the final model can be adapted to other types of datasets, Schulz *et al.* (2021).

The above-mentioned articles are summarized in table 7 with data from authors, year, parents of the first author, number of citations and main algorithm mentioned in the body of the article; articles related to research question 6.

Author	Year	Database	Country	Citations	Algorithm
[Mazzia et al. 2020]	2020	MDPI	Italy	23	KM
[Abdellah and Thangadurai	2021	IEEE	Sudan	62	CNN
2021]					
[Gumma et al. 2020]	2019	Tandfonline	USA		RF
[Tufail et al. 2021a]	2021	IEEE	Pakistan		SVM
[Shafi et al. 2020]	2020	IEEE	Pakistan	6	SVM, NB
[Aneece and Thenkabail	2021	MDPI	USA		WekaXMeans,
2021]					SVM,
					RF, NB
[Worrall et al. 2021]	2021	MDPI	USA		ANN, LSTMM
[Yang and Cho 2021]	2021	MDPI	Korea	57	MSC, DBSCAN,
					KM
[Barnes et al. 2021]	2021	MDPI	USA		RF
[Najafi et al. 2021]	2021	MDPI	Iran	7	ANN, SVM
[Xie et al. 2021]	2021	MDPI	China	7	SVR, RF
[Momm et al. 2020]	2020	MDPI	USA	7	RF
[Yan et al. 2021]	2021	ScienceDirect	China	1	CART

Table 7. Authors, year, electronic database, citations, algorithm



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

[Ju et al. 2021]	2021	ScienceDirect	South		SSAE, SVM, RF,
			Korea		DT,
					CNN, LSTMM
[Schulz et al. 2021]	2021	ScienceDirect	Germany	1	RF

3.6. RQ6 - Is it possible to elaborate on a basic taxonomy?

After collecting the articles and studying them, we investigate what is being used and pointed out by the researchers, which allowed the elaboration of a primary taxonomy presented in Figure 3. This taxonomy allows for analysis and comparisons with the latest machine learning application in agriculture. First, we present the main crops dealt with in the articles, indicating that they are mostly export commodities. Next, we indicate the cloud technologies used. Finally, we indicate that remote sensing is extensively explored in several articles. The algorithms could have been elaborated and grouped by family category, but we decided to list them here as the authors cite them in the articles. The taxonomy was designed from the collected articles; we can observe five significant ramifications (applicability, cultures and technologies, and machine learning); machine learning was subdivided into supervised and unsupervised algorithms. The technologies were subdivided into cloud, UAV, IoT, and remote sensing; in UAV, we have drones that perform data collection via aero photogrammetry, in Cloud Google Earth stands out; in IoT, the most popular microcontroller is the Raspberry Pi; in remote sensing, the most cited satellites are Sentinel 1 and 2.



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva



Figure 3. Basic taxonomy

3.7. RQ7 - What other bibliometric data can be obtained?

Several algorithms are mentioned during the work, we list the 10 most identified by the authors, they are by quantities: random forest (RF) - 40, support vector machines (SVM) - 25, artificial neural network (ANN) - 18, decision tree (DT) - 12, K nearest neighbor (KNN) - 11, convolutional neural network (CNN) 9, support vector regression (SVR) - 7, K-means (KM) 5, long short term memory model (LSTMM) - 5, naive Bayes (NB) - 5, see Figure 4. In addition to the technologies mentioned in question 6, we can see satellites mentioned in other articles with another focus, but if we add them all together, the quantities are: Landsat - 5, Sentinel-1 - 7, Sentinel-2 11, Gaofen-1/2 - 4, Earth Observing - 3, Worldview-2/3 - 2 and Terra - 3, see Figure 5.

Also, nine articles cited the use of technologies such as IoT, Arduino and Raspberry Pi. Eight articles cited the use of UAV (Unmanned Aerial Vehicle). Fifteen articles addressed topics related to Agriculture 4.0. The most cited crops are banana, barley, carrot, corn, potato, red pepper, rice, soy, sugar cane, sunflower, tea, tobacco, wheat; it is important to note that soybeans, corn, and wheat are



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

high-value agricultural commodities for export. Also, it was considered the country of the first author; then we ranked the top 10 countries with the highest number of publications: China - 16, India - 14, USA - 13, Brazil - 4, Germany - 3, Korea - 3, Pakistan - 3, Australia - 2, Canada - 2, Greece - 2; see Figure 6.





Figure 4. Top 10 most cited algorithm



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva







3.8. Overview of word cloud

We present an overview of various keywords and their relationship and limitations of this search. The software used to build bibliometric networks was VOSviewer (Waltman *et al.* 2010) and (van Eck; Waltman, 2009). The text was extracted from the title and abstract from a .RIS file. The references of each article were used in. RIS format to feed the Vosviewer tool. Full counting was chosen as a method. Of 2175 terms, 34 meet the threshold, and seven minimum occurrences of a term were selected. The number of terms selected was 40. Figure 7 presented do word grouping in words clusters and its interrelation. The clusters detected by VOSviewer were: Cluster 1: approach, crop yield, input, machine learning, model, prediction, random forest, region, remote sensing, rice, machine learning algorithm, paper, soil, time, type; Cluster 3: accuracy, classification, crop, experiment, image, information, sentinel, support vector machines, use.



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva



Figure 7. Word cloud and its interrelationships in clusters

We can extract from Figure 7 that the term machine learning algorithm is quite strong seen in the detail of Figure 8, in the same way the term random forest can be seen separately in Figure 9.



Figure 8. Detail of potential relationships between with "machine learning algorithm"



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva



Figure 9. Detail of potential relationships between with "random forest"

4. LIMITATIONS

We tried to investigate the application of machine learning in agriculture. This article is limited to the application of machine learning in agriculture only, not addressing secondary issues such as social and other fields of artificial intelligence beyond the one addressed by the authors of the selected articles. Our focus was on applying machine learning within a previously defined temporal window. Another limitation was the issue of acquiring articles exclusively via databases of scientific portals. Thus, it is possible that some articles have not entered the search radar because we removed other databases. The extraction of data, duplicates, title, and abstract was performed with the help of a large data table and manually; in this way, as quoted by Rattan *et al.* (2013), errors can occur even due to the loss of articles; efforts were made to mitigate this possibility.

5. RESEARCH QUESTIONS: COMMENTARY AND FINDINGS

Question 1 pointed out the theme of drought, water resource management, and climate forecasting; includes the weather forecast on this issue along with the issue of drought because through the predictability of rains, farmers and government entities can carry out planning against events such as drought; several algorithms are used highlighting RF and SVM, the focus of the analyzed articles mostly deals with large crops such as (rice, wheat, potato and corn) and in cases where Landsat and Sentinel satellite images stand out, indicating the possibility of a greater number



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

of projects focusing on data collection in the field with IoT devices, treatment of incomplete climatic data series.

Question 2 pointed to the needs of farmers to fight different types of pests, from microscopic ones, insects to larger animals; several studies use satellite images to detect, via multispectral images, possible variations and diseases in crops, only one work showed the identification of larger animals and one in the identification of insects, which shows the need to expand studies and tests in fields of intelligent traps. Allied to machine learning and warning systems; only one study performed an IoT mix with alerts via messaging; Here, the crops studied are high-value commodities for export, such as soybeans and corn (among others), which indicates the authors' preference for serving these crops and to the detriment of the micro family farmer, which is also an open opportunity due to the low investment power.

Question 3 indicates a few works aimed at forecasting prices or forecasting gains with the harvest; we indicate as an open question the use of data collected in the field and the use of socioeconomic data series for the region studied with global market data.

Question 4 presents how the technology can be applied in various internal aspects of the farm, its controls, monitoring, and planning; several aspects are taken into consideration, such as safety, safety at work, analysis of financial efficiency, plant coverage, harvest prediction, classification of harvests (and grains) among others; we note the possibility of exploring the collection and integration of telemetry data from agricultural machinery and the need for expansion in communication technologies for the field, as without these the exchange of data between systems would be unfeasible, more studies are needed in these areas.

Question 5 presents several technological resources in use, mainly unmanned aerial vehicles (UAV) and satellites; Other articles deal with satellites, we group ed in question 6 the most relevant; however, it is an open field to use robots within the fields combined with cloud technologies (little explored), once again this field of study touches the issue of field connectivity.

Question 6 presents a brief taxonomy, indicating the algorithms cited in the studies, indicating the possibility of exploring several other areas such as: including economic, socio-economic data, risk analysis for production based on multiple social, pandemic, climatic, and market scenarios; in addition to data on the use of new technologies in the field such as 5G and increased use of robots in agriculture.

6. CONCLUSIONS

The current study carried out a literature review to identify the use of machine learning in agriculture (the complete list of articles is listed in Appendix 1). We answer several research questions. We identify and organize results in a structured way in the document according to the questions. We suggest several other fields of study in the open questions section. We have included a basic taxonomy indicating the identified technologies; this is important as it presents what has been



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

used with machine learning and in the years of 2019-2021, updated and recent. As for recommendations for future work, we recommend: 1) a macro model or architecture that includes the new horizons pointed out in the open questions section; 2) a service that can be pointed to a folder where there are satellite images and pointed to a database where there are series of tabular data, this service can run in the background mode and have a graphical interface for its configuration, the service output would be tables and graphics processed from the input data, this service would allow its user to choose the data source, algorithm type and output type (table or graph); 3) Analyze the use of large language models such as BERT and how these models are impacting field data analysis for farmers and agricultural problem researchers and 4) Developing a macro model or framework for applying AI in agriculture arose from the need to integrate various technologies (such as IoT, big data, machine learning, and computer vision) into a unified architecture.

7. FUNDING

This research received no funding.

8. CONFLICTS OF INTEREST

The authors declare no conflict of interest.

9. ABBREVIATIONS USED IN THIS MANUSCRIPT

AAB - Adaboost Adaptative Boosting ANN – Artificial Neural Network ARIMA – Autoregressive Integrated Moving Average BM - Bayesian models BRT – Boosted Regression Trees C5 – C5.0 algorithm CART - Classification and Regression Trees **CNN – Convolutional Neural Network** DBSCAN - Density-Based Spatial Clustering of Applications with Noise DGNN – Domain Guided Neural Network **DT** – Decision Tree ELR - Extreme Learning Regression EM – Elastic Net FDA - Flexible Discriminant Analysis GA – Genetic Algorithms GB - Gradient Boosting **GBRT** – Gradient Boosted Regression Trees



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luís Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

- GLCM Gray Level Co-occurrence Matrix
- GPR Gaussian Process Regressions
- KNN K Nearest Neighbor
- KM K-means
- LGBM Light Gradient Boosting Machine
- LLM -Large Language Models
- LSTMM Long Short Term Memory Model
- MARS Multivariate Adaptive Regression Splines
- ML Maximum Likelihood
- MLPNN Multi Layer Perceptron Neural Network
- MLR Multiple Linear Regressions
- MODIS Moderate Resolution Imaging Spectrometer
- MSC Mean-Shift Clustering
- NB Naive Bayes
- NDVI Normalized Difference Vegetation Index
- PLS Partial Least Squares
- PSO Particle Swarm Optimization
- PA Prophet Algotirthm
- RF Random Forest
- SSAE Stacked Sparse Autoencoder
- SGD Stochastic Gradient Descent
- SVC Support Vector Classification
- SVM Support Vector Machines
- SVR Support Vector Regression
- XGB XGBoost Extreme Gradient Boosting

REFERENCES

ABBAS, F.; AFZAAL, H.; FAROOQUE, A. A.; TANG, S. Crop yield prediction through proximal sensing and machine learning algorithms. **Agronomy**, v. 10, n. 7, 2020. DOI: https://doi.org/10.3390/agronomy10071046.

ABDELLAH, N. A. A.; THANGADURAI, N. Real time application of iot for the agriculture in the field along with machine learning algorithm. *In*: International Conference on Computer, Control, Electrical, and Electronics Engineering (ICCCEEE), 2021. p. 1–6. DOI:10.1109/ICCCEEE49695.2021.9429606.

ACHARYA, U.; DAIGH, A. L. M.; ODUOR, P. G. Machine learning for predicting field soil moisture using soil, crop, and nearby weather station data in the red river valley of the north. **Soil Systems**, v. 5, n. 4, 2021. DOI: https://doi.org/10.3390/soilsystems5040057.



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Sliva, Lidia Martins da Silva

ADEBIYI, M. O.; OGUNDOKUN, R. O.; ABOKHAI, A. A. Machine learning– based predictive farmland optimization and crop monitoring system. **Scientifica**, 2020. DOI: 10.1155/2020/9428281.

AGRAWAL, A.; PATEL, M.; SHARMA, A. K. Novel supervised machine learning classification technique for improve accuracy of multi-valued datasets in agriculture. *In*: 6th International Conference on Inventive Computation Technologies (ICICT), 2021. p. 1067–1070. DOI:10.1109/ICICT50816.2021.9358691.

AHISH, N.; SHASHIKALA, K.; BHARARTH, N. Automated modular data analysis and visualization system with predictive analytics using machine learning for agriculture field. **International Journal of Research in Science**, v. 5, n. 1, p. 1–3, 2019. DOI:10.24178/ijrs.2019.5.1.01.

ANEECE, I.; THENKABAIL, P. S. Classifying crop types using two generations of hyperspectral sensors (hyperion and desis) with machine learning on the cloud. **Remote Sensing,** v. 13, n. 22, 2021. DOI: https://doi.org/10.3390/rs13224704.

BALAKRISHNA, K.; MOHAMMED, F.; ULLAS, C.; HEMA, C.; SONAKSHI, S. Application of iot and machine learning in crop protection against animal intrusion. **Global Transitions Proceedings,** v. 2, n. 2, p. 169–174, 2021. International Conference on Computing System and its Applications (ICCSA-2021). DOI: https://doi.org/10.1016/j.gltp.2021.08.061.

BALNE, S. Smart agriculture using advanced machine learning algorithms. **International Journal of Innovative Research in Science, Engineering and Technology**, v. 9, n. 7, p. 6836-6840, July 2020.

BANERJEE, B. P.; SHARMA, V.; SPANGENBERG, G.; KANT, S. Machine learning regression analysis for estimation of crop emergence using multispectral uav imagery. **Remote Sensing**, v. 13, n. 15, 2021, DOI: https://doi.org/10.3390/rs13152918.

BARNES, M. L.; YODER, L.; KHODAEE, M. Detecting winter cover crops and crop residues in the midwest us using machine learning classification of thermal and optical imagery. **Remote Sensing**, v. 13, n. 10, 2021. DOI: https://doi.org/10.3390/rs13101998.

BECK, M. A.; LIU, C.-Y.; BIDINOSTI, C. P.; HENRY, C. J.; GODEE, C. M.; AJMANI, M. An embedded system for the automated generation of labeled plant images to enable machine learning applications in agriculture. **PLOS ONE**, v. 15, n. 12, p. 1–23, 2020. DOI: https://doi.org/10.1371/journal.pone.0243923.

BHIMANPALLEWAR, R. N.; NARASINGARAO, M. R. Alternative approaches of machine learning for agriculture advisory system. *In*: **10th International Conference on Cloud Computing, Data Science Engineering (Confluence)**, 2020. p. 27–31, DOI: DOI:10.1109/Confluence47617.2020.9058152.

BUDGEN, D.; BRERETON, P. Performing systematic literature reviews in software engineering. *In*: **Proceedings of the 28th International Conference on Software Engineering, ICSE '06.** 2006. p. 1051–1052, New York, NY, USA. Association for Computing Machinery, DOI: https://doi.org/10.1145/1134285.1134500.

CHATERJI, S.; DELAY, N.; EVANS, J.; MOSIER, N.; ENGEL, B.; BUCKMASTER, D.; LADISCH, M. R.; CHANDRA, R. Lattice: A vision for machine learning, data engineering, and policy considerations for digital agriculture at scale. **IEEE Open Journal of the Computer Society**, v. 2, p. 227–240, 2021. DOI: DOI 10.1109/OJCS.2021.3085846.

CHEN, Z.; GOH, H. S.; SIN, K. L.; LIM, K.; KA, N.; CHUNG, H.; LIEW, X. Y. Automated Agriculture Commodity Price Prediction System with Machine Learning Techniques. Advances in Science,



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Sliva, Lidia Martins da Silva

Technology and Engineering Systems Journal, v. 6, n. 4, p. 376–384, 2021. DOI: https://doi.org/10.48550/arXiv.2106.12747.

CHINNAIYAN, R.; BALACHANDAR, S. Reliable administration framework of drones and iot sensors in agriculture farmstead using blockchain and smart contracts. *In*: **Proceedings of the 2020 2nd International Conference on Big Data Engineering and Technology**, **BDET**. 2020. p. 106–111, New York, NY, USA. Association for Computing Machinery, DOI: https://doi.org/10.1145/3378904.3378918.

CRISOSTOMO DE CASTRO FILHO, H.: ABÍLIO DE CARVALHO JUNIOR, O.: FERREIRA DE CARVALHO, O. L.; POZZOBON DE BEM, P.; DOS SANTOS DE MOURA, R.; OLINO DE ALBUQUERQUE, A.; ROSA SILVA, C.; GUIMARAES FERREIRA, P. H.; FONTES GUIMARÃES, R.; TRANCOSO GOMES, R. A. Rice crop detection using lstm, bi-lstm, and machine learning models from sentinel-1 time series. Remote Sensing, ν. 12, n. 16. 2020. DOI: https://doi.org/10.3390/rs12162655.

DANNER, M.; BERGER, K.; WOCHER, M.; MAUSER, W.; HANK, T. Efficient RTM-based training of machine learning regression algorithms to quantify biophysical biochemical traits of agricultural crops. **ISPRS Journal of Photogrammetry and Remote Sensing**, v. 173, n. 278–296, 2021. DOI: https://doi.org/10.1016/j.isprsjprs.2021.01.017.

DASH, R.; DASH, D. K.; AND BISWAL, G. Classification of crop based on macronutrients and weather data using machine learning techniques. **Results in Engineering**, v. 9, p. 100203, 2021. DOI: DOI:10.1016/j.rineng.2021.100203.

DING, Y.; ZHANG, H.; WANG, Z.; XIE, Q.; WANG, Y.; LIU, L.; HALL, C. C. A comparison of estimating crop residue cover from sentinel-2 data using empirical regressions and machine learning methods. **Remote Sensing**, v. 12, n. 9, 2020. DOI: https://doi.org/10.3390/rs12091470.

DUBOIS, A.; TEYTAUD, F.; VEREL, S. Short term soil moisture forecasts for potato crop farming: A machine learning approach. **Computers and Electronics in Agriculture**, v. 180, p. 105902, 2021. DOI: DOI: 10.1016/j.compag.2020.105902.

FENG, P.; WANG, B.; LIU, D. L.; YU, Q. Machine learning-based integration of remotely-sensed drought factors can improve the estimation of agricultural drought in south-eastern australia. **Agricultural Systems**, v. 173, p. 303–316, 2019. DOI: 10.1016/j.agsy.2019.03.015.

FENU, G.; MALLOCI, F. M. An application of machine learning technique in forecasting crop disease. *In* **Proceedings of the 2019 3rd International Conference on Big Data Research, ICBDR** 2019. p. 76–82, New York, NY, USA. Association for Computing Machinery, DOI: https://doi.org/10.1145/3372454.3372474.

FEYISA, G. L.; PALAO, L. K.; NELSON, A.; GUMMA, M. K.; PALIWAL, A.; WIN, K. T.; NGE, K. H.; JOHNSON, D. E. Characterizing and mapping cropping patterns in a complex agro-ecosystem: An iterative participatory mapping procedure using machine learning algorithms and modis vegetation indices. **Computers and Electronics in Agriculture**, v. 175, p. 105595, 2020. DOI: https://doi.org/10.1016/j.compag.2020.105595.

FOLBERTH, C.; BAKLANOV, A.; BALKOVIC, J.; SKALSK^{*} Y, R.; KHABAROV, N.; OBERSTEINER, M. Spatio-temporal downscaling of gridded crop model yield estimates based on machine learning. **Agricultural and Forest Meteorology**, v. 264, p. 1–15, 2019. DOI: https://doi.org/10.1016/j.agrformet.2018.09.021.



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

GARG, S.; PUNDIR, P.; JINDAL, H.; SAINI, H.; GARG, S. Towards a multimodal system for precision agriculture using iot and machine learning. **Computer Science**, 2021. DOI: DOI:10.48550/arXiv.2107.04895.

GONG, D.; HAO, W.; GAO, L.; FENG, Y.; CUI, N. Extreme learning machine for reference crop evapotranspiration estimation: Model optimization and spatiotemporal assessment across different climates in china. **Computers and Electronics in Agriculture**, v. 187, p. 106294, 2021. DOI: 10.1016/j.compag.2021.106294.

GUMMA, M. K.; THENKABAIL, P. S.; TELUGUNTLA, P. G.; OLIPHANT, A.; XIONG, J.; GIRI, C.; PYLA, V.; DIXIT, S.; WHITBREAD, A. M. Agricultural cropland extent and areas of south asia derived using landsat satellite 30-m time-series big-data using random forest machine learning algorithms on the google earth engine cloud. **GIScience & Remote Sensing**, v. 57, n. 3, p. 302–322, 2020. DOI: https://doi.org/10.1080/15481603.2019.1690780.

HU, B.; XUE, J.; ZHOU, Y.; SHAO, S.; FU, Z.; LI, Y.; CHEN, S.; QI, L.; SHI, Z. Modelling bioaccumulation of heavy metals in soil-crop ecosystems and identifying its controlling factors using machine learning. **Environmental Pollution**, v. 262, p. 114308, 2020. DOI: https://doi.org/10.1016/j.envpol.2020.114308.

IATROU, M.; KARYDAS, C.; IATROU, G.; PITSIORLAS, I.; ASCHONITIS, V.; RAPTIS, I.; MPETAS, S.; KRAVVAS, K.; MOURELATOS, S. Topdressing nitrogen demand prediction in rice crop using machine learning systems. **Agriculture**, v. 11, n. 4, 2021. DOI: https://doi.org/10.3390/agriculture11040312.

JAGTAP, S. T.; PHASINAM, K.; KASSANUK, T.; JHA, S. S.; GHOSH, T.; THAKAR, C. M. Towards application of various machine learning techniques in agriculture. **Materials Today: Proceedings**, 2021. DOI:10.1016/j.matpr.2021.06.236.

JU, S.; LIM, H.; MA, J. W.; KIM, S.; LEE, K.; ZHAO, S.; HEO, J. Optimal county-level crop yield prediction using modis-based variables and weather data: A comparative study on machine learning models. **Agricultural and Forest Meteorology**, v. 307, p. 108530, 2021. DOI: 10.1016/j.agrformet.2021.108530.

KASINATHAN, T.; SINGARAJU, D.; UYYALA, S. R. Insect classification and detection in field crops using modern machine learning techniques. **Information Processing in Agriculture**, v. 8, n. 3, p. 446–457, 2021. DOI: https://doi.org/10.1016/j.inpa.2020.09.006.

KETCHUM, D.; JENCSO, K.; MANETA, M. P.; MELTON, F.; JONES, M. O.; HUNTINGTON, J. Irrmapper: A machine learning approach for high resolution mapping of irrigated agriculture across the western u.s. **Remote Sensing**, v. 12, n. 14, 2020. DOI: https://doi.org/10.3390/rs12142328.

KGANYAGO, M.; MHANGARA, P.; ADJORLOLO, C. Estimating crop biophysical parameters using machine learning algorithms and sentinel-2 imagery. **Remote Sensing**, v. 13, n. 21, 2021. DOI: https://doi.org/10.3390/rs13214314.

KWAK, G.-H.; PARK, N.-W. Impact of texture information on crop classification with machine learning and UAV images. **Applied Sciences**, v. 9, n. 4, 2021. DOI: https://doi.org/10.3390/app9040643.

LI, Z.; ZHANG, Z.; ZHANG, L. Improving regional wheat drought risk assessment for insurance application by integrating scenario-driven crop model, machine learning, and satellite data. **Agricultural Systems**, V. 191, p. 103141, 2021. DOI: https://doi.org/10.1016/j.agsy.2021.103141.



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

LIU, L.; ZHAN, X. Analysis of financing efficiency of chinese agricultural listed companies based on machine learning. **Complexity,** v. 2019, p. 9190273, 2019. DOI: https://doi.org/10.1155/2019/9190273.

MAIMAITIJIANG, M.; SAGAN, V.; SIDIKE, P.; DALOYE, A. M.; ERKBOL, H.; FRITSCHI, F. B. (2020). Crop monitoring using satellite/UAV data fusion and machine learning. **Remote Sensing**, v. 12, n. 9, 2019, DOI: https://doi.org/10.3390/rs12091357.

MANRIQUE-SILUPU, J.; CAMPOS, J. C.; PAIVA, E.; IPANAQUE, W. Thrips incidence prediction in organic banana crop with machine learning. **Heliyon**, p. e08575, 2021. http://dx.doi.org/10.1016/j.heliyon.2021.e08575.

MAPONYA, M. G.; VAN NIEKERK, A.; MASHIMBYE, Z. E. Pre-harvest classification of crop types using a sentinel-2 time-series and machine learning. **Computers and Electronics in Agriculture**, v. 169, p. 105164, 2020. http://dx.doi.org/10.1016/j.compag.2019.105164.

MATEO-SANCHIS, A.; PILES, M.; AMOROS-L´ OPEZ, J.; MU´ NOZ-MAR~ 1, J.; ADSUARA, J. E.; MORENO-MART'INEZ, Alvaro; CAMPS-VALLS, G. Learning main drivers of crop progress and failure in europe with interpretable machine learning. International Journal of Applied Earth Observation and Geoinformation, v. 104, p. 102574, 2021. DOI: https://doi.org/10.1016/j.jag.2021.102574.

MAZZIA, V.; COMBA, L.; KHALIQ, A.; CHIABERGE, M.; GAY, P. UAV and machine learning based refinement of a satellite-driven vegetation index for precision agriculture. **Sensors,** v. 20, n. 9, 2020. DOI: https://doi.org/10.3390/s20092530.

MIAN, P.; CONTE, T.; NATALI, A.; BIOLCHINI, J.; TRAVASSOS, G. A systematic review process to software engineering - technical report rt-es 679/05. [*S. l.:* s. n.], 2005.

MOMM, H. G.; ELKADIRI, R.; PORTER, W. Crop-type classification for longterm modeling: An integrated remote sensing and machine learning approach. **Remote Sensing**, v. 12, n. 3, 2020. DOI: https://doi.org/10.3390/rs12030449.

MOUHSSINE, R.; OTMAN, A.; KHATIR HAIMOUDI, E. Performance analysis of machine learning techniques for smart agriculture: Comparison of supervised classification approaches. **International Journal of Advanced Computer Science and Applications**, v. 11, n. 3, 2020. DOI: https://dx.doi.org/10.14569/IJACSA.2020.0110377.

MOUMNI, A.; LAHROUNI, A. Machine learning-based classification for croptype mapping using the fusion of high-resolution satellite imagery in a semiarid area. **Scientifica**, v. 2021, p. 8810279, 2021. DOI: https://doi.org/10.1155/2021/8810279.

MUNIASAMY, A. Machine learning for smart farming: A focus on desert agriculture. *In*: International Conference on Computing and Information Technology (ICCIT-1441), 2020. p. 1–5, DOI: 10.1109/ICCIT-144147971.2020.9213759.

MUTHONI, F.; THIERFELDER, C.; MUDERERI, B.; MANDA, J.; BEKUNDA, M.; HOESCHLEZELEDON, I. Machine learning model accurately predict maize grain yields in conservation agriculture systems in southern africa. *In*: **9th International Conference on Agro-Geoinformatics (Agro-Geoinformatics),** 2021. p. 1–5, DOI: http://dx.doi.org/10.1109/Agro-Geoinformatics50104.2021.9530335.

NAJAFI, P.; FEIZIZADEH, B.; NAVID, H. A comparative approach of fuzzy object based image analysis and machine learning techniques which are applied to crop residue cover mapping by using



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

sentinel-2 satellite and uav imagery. **Remote Sensing,** v. 13, n. 5, 2021. DOI: https://doi.org/10.3390/rs13050937.

PANT, J.; PANT, R.; KUMAR SINGH, M.; PRATAP SINGH, D.; PANT, H. Analysis of agricultural crop yield prediction using statistical techniques of machine learning. **Materials Today: Proceedings,** v. 46, p. 10922–10926, 2021. DOI: 10.1016/j.matpr.2021.01.948.

PAUDEL, D.; BOOGAARD, H.; DE WIT, A.; JANSSEN, S.; OSINGA, S.; PYLIANIDIS, C.; ATHANASIADIS, I. N. Machine learning for large-scale crop yield forecasting. **Agricultural Systems**, v. 187, p. 103016, 2021. DOI: https://doi.org/10.1016/j.agsy.2020.103016.

PEPPES, N.; DASKALAKIS, E.; ALEXAKIS, T.; ADAMOPOULOU, E.; DEMESTICHAS, K. Performance of machine learning-based multi-model voting ensemble methods for network threat detection in agriculture 4.0. **Sensors**, v. 1, n. 22, 2021. DOI: https://doi.org/10.3390/s21227475.

PREMACHANDRA, J. S. A. N. W.; KUMARA, P. P. N. V. A novel approach for weather prediction for agriculture in sri lanka using machine learning techniques. *In*: International Research Conference on Smart Computing and Systems Engineering (SCSE), v 4, p. 182–189, 2021. DOI:10.1109/SCSE53661.2021.9568319.

PRODHAN, F. A.; ZHANG, J.; PANGALI SHARMA, T. P.; NANZAD, L.; ZHANG, D.; SEKA, A. M.; AHMED, N.; HASAN, S. S.; HOQUE, M. Z.; MOHANA, H. P. Projection of future drought and its impact on simulated crop yield over south asia using ensemble machine learning approach. **Science of The Total Environment**, p. 151029, 2021. DOI: 10.1016/j.scitotenv.2021.151029.

RADHAKRISHNAN, S. An improved machine learning algorithm for predicting blast disease in paddy crop. **Materials Today: Proceedings**, v. 33, p. 682–686, 2020. DOI: https://doi.org/10.1016/j.matpr.2020.05.802.

RAHMATI, O.; FALAH, F.; DAYAL, K. S.; DEO, R. C.; MOHAMMADI, F.; BIGGS, T.; MOGHADDAM, D. D.; NAGHIBI, S. A.; BUI, D. T. Machine learning approaches for spatial modeling of agricultural droughts in the south-east region of queensland australia. **Science of The Total Environment**, v. 699, p. 134230, 2020. DOI: 10.1016/j.scitotenv.2019.134230.

RATTAN, D.; BHATIA, R.; SINGH, M. Software clone detection: A systematic review. **Information** and **Software Technology**, v. 55, n. 7, p. 1165–1199, 2020. DOI: https://doi.org/10.1016/j.infsof.2013.01.008.

SCHULZ, C.; HOLTGRAVE, A.-K.; KLEINSCHMIT, B. Large-scale winter catch crop monitoring with sentinel-2 time series and machine learning–an alternative to on-site controls? **Computers and Electronics in Agriculture**, v. 186, p. 106173, 2020. DOI: 10.1016/j.compag.2021.106173.

SCHWALBERT, R. A.; AMADO, T.; CORASSA, G.; POTT, L. P.; PRASAD, P.; CIAMPITTI, I. A. Satellite-based soybean yield forecast: Integrating machine learning and weather data for improving crop yield prediction in southern brazil. **Agricultural and Forest Meteorology**, v. 284, p. 107886, 2020. DOI: https://doi.org/10.1016/j.agrformet.2019.107886.

SCOTT, E.; HIRABAYASHI, L.; LEVENSTEIN, A.; KRUPA, N.; JENKINS, P. The development of a machine learning algorithm to identify occupational injuries in agriculture using pre-hospital care reports. **Health Information Science and Systems**, v. 9, n. 1, p. 31, 2020. DOI: https://doi.org/10.1007/s13755-021-00161-9.

SHAFI, U.; MUMTAZ, R.; IQBAL, N.; ZAIDI, S. M. H.; ZAIDI, S. A. R.; HUSSAIN, I.; MAHMOOD, Z. A multi-modal approach for crop health mapping using low altitude remote sensing, internet of things



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

(iot) and machine learning. **IEEE Access**, v. 8, n. 112708–112724, 2020. DOI:10.1109/ACCESS.2020.3002948.

SHUKLA, R.; DUBEY, G.; MALIK, P.; SINDHWANI, N.; ANAND, R.; DAHIYA, A.; YADAV, V. Detecting crop health using machine learning techniques in smart agriculture system. **Journal of Scientific and Industrial Research (JSIR)**, v. 80, p. 699–706, 2021.

SKAWSANG, S.; NAGAI, M. K.; TRIPATHI, N.; SONI, P. Predicting rice pest population occurrence with satellite-derived crop phenology, ground meteorological observation, and machine learning: A case study for the central plain of thailand. **Applied Sciences**, v. 9, n. 22, 2021. DOI: https://doi.org/10.3390/app9224846.

SOSA, L.; JUSTEL, A.; MOLINA, Detection of crop hail damage with a machine learning algorithm using time series of remote sensing data. **Agronomy,** v. 11, n. 10, 2021. DOI: https://doi.org/10.3390/agronomy11102078.

TANG, Z.; WANG, H.; LI, X.; LI, X.; CAI, W.; HAN, C. An object-based approach for mapping crop coverage using multiscale weighted and machine learning methods. *In:* **IEEE Journal of Selected Topics in Applied Earth Observations and Remote Sensing**, v. 13, p. 1700–1713, 2020. DOI:10.1109/JSTARS.2020.2983439.

TAO, Y.; YOU, F. Prediction of cover crop adoption through machine learning models using satellitederived data. **IFAC-PapersOnLine**, v. 52, n. 30, p. 137–142, 2019. DOI: https://doi.org/10.1016/j.ifacol.2019.12.511.

TOWETT, E. K.; DRAKE, L. B.; ACQUAH, G. E.; HAEFELE, S. M.; MCGRATH, S. P.; SHEPHERD, K. D. Comprehensive nutrient analysis in agricultural organic amendments through non-destructive assays using machine learning. **PLOS ONE**, v. 15, n. 12, p. 1–17, 2019. DOI: https://doi.org/10.1371/journal.pone.0242821.

TUFAIL, M.; IQBAL, J.; TIWANA, M. I.; ALAM, M. S.; KHAN, Z. A.; KHAN, M. T. Identification of tobacco crop based on machine learning for a precision agricultural sprayer. **IEEE Access**, v. 9, p. 23814–23825, 2021a. DOI: 10.1109/ACCESS.2021.3056577.

TUFAIL, R.; AHMAD, A.; JAVED, M. A.; AHMAD, S. R. A machine learning approach for accurate crop type mapping using combined SAR and optical time series data. **Advances in Space Research**, 2021b. DOI: 10.1016/j.asr.2021.09.019.

VAN ECK, N. J.; WALTMAN, L. Software survey: Vosviewer, a computer program for bibliometric mapping. **Scientometrics**, v. 84, p. 523-538, 2009. DOI: https://doi.org/10.1007/s11192-009-0146-3.

WALEED, M.; UM, T.-W.; KAMAL, T.; USMAN, S. M. Classification of agriculture farm machinery using machine learning and internet of things. **Symmetry,** v. 13, n. 3, 2021. DOI: https://doi.org/10.3390/sym13030403.

WALTMAN, L.; VAN ECK, N. J.; NOYONS, E. C. A unified approach to mapping and clustering of bibliometric networks. **Journal of Informetrics**, v. 4, n. 4, p. 629–635, 2010. DOI: https://doi.org/10.1016/j.joi.2010.07.002.

WANG, C.; LIU, S.; ZHENG, H.; HU, H.; SONG, L. Application of machine learning in comprehensive evaluation of agricultural high-tech. *In*: **Proceedings of the 3rd International Conference on Computer Science and Application Engineering, CSAE,** 2019. DOI: https://doi.org/10.3390/s21113758.



MACHINE LEARNING ALGORITHMS IN AGRICULTURE: A LITERATURE REVIEW ON CLIMATE AND PRICE PREDICTION, PEST AND DISEASE DETECTION, AND PRODUCTION MONITORING Emiliano Soares Monteiro, Rodrigo da Rosa Righi, Antônio Marcos Alberti, Sandro José Rigo, Jorge Luis Victoria Barbosa, Perla Haydee da Silva, Lidia Martins da Silva

WANG, S.; GUAN, K.; WANG, Z.; AINSWORTH, E. A.; ZHENG, T.; TOWNSEND, P. A.; LIU, N.; NAFZIGER, E.; MASTERS, M. D.; LI, K.; WU, G.; JIANG, C. Airborne hyperspectral imaging of nitrogen deficiency on crop traits and yield of maize by machine learning and radiative transfer modeling. **International Journal of Applied Earth Observation and Geoinformation**, v. 105, p. 102617, 2021. DOI: https://doi.org/10.1016/j.jag.2021.102617.

WEI, M. C. F.; MALDANER, L. F.; OTTONI, P. M. N.; MOLIN, J. P. Carrot yield mapping: A precision agriculture approach based on machine learning. **AI**, v. 1, n. 2, p. 229-241, 2020. DOI: DOI: 10.3390/ai1020015.

WOLANIN, A.; CAMPS-VALLS, G.; GOMEZ-CHOVA, L.; MATEO-GARCIA, G.; VAN DER TOL, C.; ZHANG, Y.; GUANTER, L. Estimating crop primary productivity with sentinel2 and landsat 8 using machine learning methods trained with radiative transfer simulations. **Remote Sensing of Environment**, v. 225, p. 441–457, DOI: 10.3390/ai1020015. DOI:10.48550/arXiv.2012.1210.

WORRALL, G.; RANGARAJAN, A.; JUDGE, J. Domain-guided machine learning for remotely sensed in-season crop growth estimation. **Remote Sensing**, v. 13, n. 22, 2021. DOI:10.3390/rs13224605.

XAVIER, L. C. P.; SILVA, S. M. O. D.; CARVALHO, T. M. N.; PONTES FILHO, J. D.; SOUZA FILHO, F. D. A. D. Use of machine learning in evaluation of drought perception in irrigated agriculture: The case of an irrigated perimeter in Brazil. **Water**, v. 12, n. 6, 2020. DOI: https://doi.org/10.3390/w12061546.

XIE, Q.; WANG, J.; LOPEZ-SANCHEZ, J. M.; PENG, X.; LIAO, C.; SHANG, J.; ZHU, J.; FU, H.; BALLESTER-BERMAN, J. D. Crop height estimation of corn from multi-year radarsat-2 polarimetric observables using machine learning. **Remote Sensing**, v. 13, n. 3, 2020. DOI:10.3390/rs13030392.

YAMAC, S. S.; TODOROVIC, M. Estimation of daily potato crop evapotranspiration using three different machine learning algorithms and four scenarios of available meteorological data. **Agricultural Water Management**, v. 228, p. 105875, 2020.

YAN, S.; YAO, X.; ZHU, D.; LIU, D.; ZHANG, L.; YU, G.; GAO, B.; YANG, J.; YUN, W. Large-scale crop mapping from multi-source optical satellite imageries using machine learning with discrete grids. **International Journal of Applied Earth Observation and Geoinformation**, v. 103, p. 102485, 2020.

YANG, M.; CHO, S.-I. High-resolution 3d crop reconstruction and automatic analysis of phenotyping index using machine learning. **Agriculture**, v. 11, n. 10, 2021. DOI:10.3390/agriculture11101010.

YANG, N.; LIU, D.; FENG, Q.; XIONG, Q.; ZHANG, L.; REN, T.; ZHAO, Y.; ZHU, D.; HUANG, J. Large-scale crop mapping based on machine learning and parallel computation with grids. **Remote Sensing**, v. 11, n. 12, 2019. DOI:10.3390/rs11121500.

YUAN, C. Z.; SAN, W. W.; LEONG, T. W. Determining optimal lag time selection function with novel machine learning strategies for better agricultural commodity prices forecasting in Malaysia. *In*: **Proceedings of the 2020 2nd International Conference on Information Technology and Computer Communications, ITCC.** 2020. p. 37–42. DOI:10.1145/3417473.3417480.

ZHANG, J.; HE, Y.; YUAN, L.; LIU, P.; ZHOU, X.; HUANG, Y. Machine learningbased spectral library for crop classification and status monitoring. **Agronomy**, v. 9, n. 9, 2019. DOI: https://doi.org/10.3390/agronomy9090496.